Author’s response to reviews:

Title: CT Imaging History for Patients Presenting to the ED with Renal Colic—Evidence from a Multi-Hospital Database

Authors:
Emily Schmid (emilyschmid8@gmail.com)
Kimberly Leeson (kleeson@gmail.com)
K Xu (ktomxu@gmail.com)
Peter Richman (prichmanmdmba@gmail.com)
Crystal Nwosu (crystal.nwous@outlook.com)
Lynn Carrasco (lynn.carrasco@christushealth.org)

Version: 2 Date: 26 Jan 2019

Author’s response to reviews:

We have moved key words to below the abstract and made sure the abstracts on the paper and within the submission system match....thanks so much, Pete